Author's response to reviews

Title: Serum homocysteine levels are decreased in levothyroxine-treated women with autoimmune thyroiditis.

Authors:

Maciej Owecki (mowecki@ump.edu.pl)
Jolanta Dorszewska (dorszewskaj@ump.edu.pl)
Nadia Sawicka (nyha@o2.pl)
Anna Oczkowska (annapol2@autograf.pl)
Michał Owecki (mkowecki@ump.edu.pl)
Michał Michalak (michal@ump.edu.pl)
Jakub Fischbach (jakub@fischbach.pl)
Wojciech Kozubski (wkozubski@ump.edu.pl)
Marek Ruchała (mruchala@ump.edu.pl)

Version: 2 Date: 26 November 2013

Author's response to reviews:

We are very grateful for prompt revision. We provided all changes required by the Editor:

1. abstract is corrected to be consistent with the abstract provided in the main text
2. There is no person who should be mentioned in Acknowledgement therefore this section was skipped. We added "Acknowledgement None"
3. We provided Competing interests section and developed the Funding section.
   "The total cost of measurements (immunoassays and high-performance liquid chromatography techniques) were founded by Poznan University of Medical Sciences Internal Funding. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of manuscript. The authors have not been paid to carry out the study and write the manuscript."